Author’s response to reviews

Title: The Outcome Of Kidney Transplants With Multiple Renal Arteries

Authors:

Dr Cagatay Aydin (caga33@superonline.com)
Ibrahim Berber (iberber@hotmail.com)
Gulum Altaca (gulumaltaca@superonline.com)
Bulent Yigit (b.vigit@excite.com)
Izzet Titiz (titizltd@superonline.com)

Version: 3 Date: 29 Jan 2004

1. In the routine follow-up protocol of our patients, we do not use nuclear medicine scan for the assessment of renal function at our institution. So, we could not do any revision according to referee 1 Matthew Cooper’s comments.

2. According to the comments of referee 2 Bedeir Ali-El-Dein we did these revisions:
   - We made a table to summarize the results for the purpose of a better clarification.
   - We also included a survival chart for graft and patient survival.
   - All of the spelling errors pointed in the minor essential revisions were corrected.
   - P values were stated as exact figures.
   - The details of the reconstructive techniques were transferred from the section of RESULTS to the section of PATIENTS and METHODS.

In this study, we reviewed the patients received allografts with multiple renal arteries. A more detailed study comparing the outcome of allografts with multiple and single renal artery is continuing at our institution.